Author’s response to reviews

Title: Modular stem in total hip arthroplasty for patients with trochanter valgus deformity: surgical technique and case series

Authors:

Xiangpeng Kong (18810999609@163.com)
Wei Chai (chaiweiguanjie@sina.com)
Minzhi Yang (yangminzhi1995@163.com)
Alvin Ong (alvin.ong@rothmaninstitute.com)
Jiying Chen (361782992@qq.com)
Yan Wang (wangyanguanjie@sina.com)
Yonggang Zhou (zhouyg301@sina.com)

Version: 2 Date: 09 Feb 2020

Author’s response to reviews:

Dear editor and reviewers,

I am very glad to receive your email. Thanks for your kind comments. The current version of the manuscript has been clean and conformed to the journal style.

In response to the reviewer's comments, the responses are as follows,

Given that hip arthritis combining with trochanter valgus deformity were relative rare (12 cases in 9 years), the suitable control cases could hardly be found to conduct the case-control study. No comparison to other prostheses or other surgical methods would inevitably affect the persuasiveness of the study on technical notes. Furthermore, because it was a retrospective case-series study, we had no need of the prospective ethics approval. In the future, multicenter randomized controlled trial will be performed to further evaluate its safety and effectiveness. We have added the explanation into the limitation section.
We have improved the manuscript and revised the “Consent for publication” section (written consent of publication)

We have included the individual contributions of each author in the “Declarations” section.

We have upload the figures as the separate files and clean the manuscript.

We have correct the wrong spelling “toll” to ”tool”.

Thank you again for your support of our study. If there is anything I need to do, please contact me without hesitation. We will try our best to modify it to meet your requirements.

Best wishes,

The authors.